Author's response to reviews

Title: The spectrum of thyroid dysfunction in an Australian hepatitis C population treated with combination Interferon-alpha-2beta and Ribavirin

Authors:

Huy A Tran (huy.tran@hnehealth.nsw.gov.au)
Tracey L Jones (tracey.jones@hnehealth.nsw.gov.au)
Robert G Batey (robert.batey@hnehealth.nsw.gov.au)

Version: 4 Date: 6 October 2005

Author's response to reviews:

No changes have been made to this second revised manuscript other than those relating formatting changes requested by the Biomed Central Editorial team.